Author's response to reviews

Title: Evaluation of the acceptability and usefulness of an information website for caregivers of people with bipolar disorder

Authors:

Lesley Berk (l.berk@deakin.edu.au)
Michael Berk (mikebe@barwonhealth.org.au)
Seetal Dodd (settal@barwonhealth.org.au)
Claire M Kelly (claire.m.kelly@gmail.com)
Stefan Cvetovski (stefanc@unimelb.edu.au)
Anthony F Jorm (ajorm@unimelb.edu.au)

Version: 6 Date: 3 June 2013

Author's response to reviews: see over
Covering letter to address editors' comments

Title: Evaluation of the acceptance and usefulness of an information website for caregivers of people with bipolar disorder

Authors:
Lesley Berk
Michael Berk
Seetal Dodd
Claire Kelly
Stefan Cvetkovski
Anthony Francis Jorm

Version 3: Date 1 June 2013

Thank you for checking the manuscript so carefully. Please find our responses to the editorial points you highlighted below:

(1) **Please include an abbreviation section**
A list of abbreviations has been added on page 24

(2) **For the author contribution author CMK should be indicated and not CD.**
CD was incorrect and we have replaced with CK referring to Claire Kelly in the section on author contributions on page 25.

(3) **Also please indicate that all authors read and approved final manuscript.**
This has been indicated in the section on author contributions on page 25.

As far as we are aware the revised manuscript conforms to the journal style.